Author’s response to reviews

Title: Pulmonary Arterial Hypertension in a Patient Treated with Dasatinib

Authors:
Andris Skride (andris.skride@gmail.com)
Matiss Sablinskis (matiss.sablinskis@gmail.com)
Kristaps Sablinskis (kristaps.sablinskis@gmail.com)
Krista Lesina (krista.lesina@gmail.com)
Aivars Lejnieks (lejnieks@latnet.lv)
Sandra Lejnieze (drsandra.lejnieze@gmail.com)

Version: 2 Date: 02 Oct 2017

Author’s response to reviews:
Introduction section has been updated accordingly.
We strongly believe that all relevant medical history has been provided, and going into details such as social history or family history is unnecessary and would only clutter the case report.
Vital signs on first admission have been updated.
Urinalysis was done, as well as CBC, complete biochemistry, but since there were no significant abnormalities, we decided to not include the results in the paper, because it would clutter it.
Information about follow-up is already provided, as the patient had another catheterization a month later, and has not been experiencing recurring symptoms, and is still using sildenafil therapy.
We added brief summary on why this case should be reported.